Author's response to reviews

Title: Two patients with intestinal failure requiring home parenteral nutrition, a NOD2 mutation and tuberculous lymphadenitis

Authors:

Holger Schäffler (holger.schaeffler@med.uni-rostock.de)
Matthias Teufel (matthias.teufel@med.uni-tuebingen.de)
Sabrina Fleischer (sabrina.fleischer@med.uni-tuebingen.de)
Chih-Jen Hsieh (chih-jen.hsieh@med.uni-tuebingen.de)
Georg Lamprecht (georg.lamprecht@med.uni-rostock.de)

Version: 7 Date: 20 February 2014

Author's response to reviews: see over
Re-Submission of the manuscript

Two patients with intestinal failure requiring home parenteral nutrition, a NOD2 mutation and tuberculous lymphadenitis

Dear Professor Trautwein,

Thank you for processing our manuscript.

Prof. Julia-Stefanie Frick was added as a co-author. She expanded the Materials and Methods section for the performed microbiological procedures. She is working in the Institute of Medical Microbiology and Hygiene, University of Tübingen, Germany.

We hope that the paper is now ready for publication.

Sincerely

[Signature]

Prof. Dr. Georg Lamprecht